Author’s response to reviews

Title: The unmet need for Emergency Obstetric Care in Tanga Region, Tanzania

Authors:

Helen Prytherch (helen.prytherch@unibas.ch)
Siriel Massawe (smassawe@muchs.ac.tz)
Rainer Kuelker (Rainer.Kuelker@t-online.de)
Claudia Hunger (Claudiahunger@aol.com)
Ferdinand Mtatifikolo (dhsp@tanga.net)
Albrecht Jahn (albrecht.jahn@urz.uni-heidelberg.de)

Version: 6 Date: 18 July 2007

Author’s response to reviews:

This revision is responding to the formatting requirements as specified in the email of 11 Juli.

All these requirements have been met. The only problem was to put all others in one line. This would have required to replace the first names by initials, which we wanted to avoid. We have now used a smaller font and we are not sure whether this is acceptable.

Many thanks to the editorial and production team!

Albrecht Jahn